\documentstyle[12pt,fleqn]{article}
\pagestyle{empty}
\setlength{\topmargin}{-1in}
\setlength{\oddsidemargin}{-.3in}
\setlength{\textheight}{9in}
\setlength{\textwidth}{7.2in}

\begin{document}


Math 105 - 4

Quiz 3
\vspace{.1in}
\begin{enumerate}

\item  Name \rule{5.in}{.3pt}

\item (10 points) Factor this polynomial completely.(All factors not just the rational ones.)
\[
x^3-3x+4x-2
\]
\vspace{3.8in}

\item (10 points)  Find a polynomial with rational coefficients that has these zeros: $-2,4+i$  (Make sure you multiply it out)





\end{enumerate}




\end{document}






